Answering Reviewers

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “One case had fetal heart rate Exceeding 200 Beats per Minute in the third trimester: Case report” (NO.91818). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in yellow and red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Reviewer #1:

1. Response to comment: (There was "...heart rate was controlled by betaloc treatment after transfer to a superior hospital...")

Response: We are very sorry for our negligence of "Betaloc" to a generic name. It has been corrected to the generic name Metoprolol and marked in yellow in the text (Page 4, line 109).

2. Response to comment: (The CARE ...the "strengths and limitations...")

Response: We have made correction according to the Reviewer’s comments (Page 6, lines 177-179).

3. Response to comment: (Please recheck the spelling of "labeolol," "QT profligation," and "sotalolol.")

Response: We are very sorry for our incorrect writing "labeolol," "QT profligation," and "sotalolol." We have re-written this part according to the Reviewer’s suggestion. It has been revised to labetalol, Long QT Syndrome, sotalol, and marked with yellow in the article.

4. Response to comment: (-Please consistently spell "flucanide" or "flukanide" throughout the manuscript.)

Response: We have re-written this section as suggested by the reviewers and it is
currently all "flucanide".

Special thanks to you for your good comments.

Reviewer #2: EDITORIAL OFFICE’S COMMENTS

1. Response to comment: (Table(s) and figure(s))
   Response: Considering the Reviewer’s suggestion, we have provided the original imaging data in the form of PPT. (Pages 10-11, lines 287-298) or (9) 91818-Figures.

2. Response to comment: (About author contribution section)
   Response: We have arranged it separately, see document ③ in (12) 91818-Supplementary Material.

3. Response to comment: (About the Core tips section)
   Response: We have made correction according to the Reviewer’s comments. (Page 1, lines 17-22) or (2) 91818-Audio Core Tip.

4. Response to comment: (The DOI number for the reference)
   Response: We have made correction according to the Reviewer’s comments. (Pages 7-10, lines 191-281)

5. Response to comment: (About Title “a”)
   Response: Change to One case had fetal heart rate Exceeding 200 Beats per Minute in the third trimester: Case report.

6. Response to comment: (About the presentation of medical records)
   Response: We have made correction according to the Reviewer’s comments. (Pages 2-4, lines 34-100)

7. Response to comment: (Original version of the first page of medical records(PDF))
   Response: See ⑤ in (12) 91818-Supplementary Material. The first page of the patient(s)’ medical record.

8. Response to comment: (CARE)
   Response: We have made correction according to the Reviewer’s comments. See to (11) 91818-CARE Checklist – 2016

9. Response to comment: (Highlights of cutting-edge research results)
   Response: We have searched Pubmed for many times, and the treatment of this type
of case is still in the empirical stage, and the treatment plan has not been standardized.

10. Response to comment: (The Revised Files)

**Response:** We are very sorry that this case does not involve (5), (8), (10), so it cannot be uploaded; The rest have been uploaded as requested.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper. And here we did not list the changes but marked in yellow in revised paper.

We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.